Author's response to reviews

Title: Statins and the risk of type 2 diabetes mellitus: cohort study using the UK Clinical Practice Research Datalink

Authors:

Ana Filipa Macedo (filipa_macedopt@hotmail.com)
Ian Douglas (Ian.Douglas@lshtm.ac.uk)
Liam Smeeth (Liam.Smeeth@lshtm.ac.uk)
Harriet Forbes (Harriet.Forbes@lshtm.ac.uk)
Shah Ebrahim (Shah.Ebrahim@lshtm.ac.uk)

Version: 5
Date: 3 July 2014

Author's response to reviews:

Changes made:
1) Title page, names, institutions, emails of all authors
2) emails match in system and paper
3) coverletter added
4) "Competing Interest, Authors' Contribution and Acknowledgement" before references list
5) line numbers added to paper, paper uploaded again
5)